Author's response to reviews

Title: The HELLP syndrome: clinical issues and management. A Review

Authors:

  Kjell Haram (kjell.haram@broadpark.no)
  Einar Svendsen (einar.svendsen@gades.uib.no)
  Ulrich Abildgaard (ulricha@ulrik.uio.no)

Version: 3  Date: 15 January 2009

Author's response to reviews:

Dear Editor,

Referring to your E-mail from January 9th, 2009 we herewith resubmit our manuscript entitled The HELLP syndrome: Clinical issues and management. A Review.

The manuscript is submitted with the tables placed behind the references. These tables are adjusted according to your requirements.

We do not believe that the references are out of order. However, table 3 has to be inserted just behind the reference to table 3 in the manuscript. The next reference in the manuscript is number 57.

We have substituted reference 18 (Williams) with reference by Barton and Sibai number 10 and the next part of the sentence is deleted, namely on page 5 “while black women with HELLP often develop this disorder in their first pregnancy (14,29)”.

Furthermore, we have performed language correction again.

If we do not succeed sending the manuscript by your electronical system, we will send the manuscript by E-mail.

We also send the same manuscript with the tables included in the text.